

Temi di Discussione

(Working Papers)

Secular stagnation, R&D, public investment and monetary policy: a global-model perspective

by Pietro Cova, Patrizio Pagano, Alessandro Notarpietro and Massimiliano Pisani





Temi di discussione

(Working papers)

Secular stagnation, R&D, public investment and monetary policy: a global-model perspective

by Pietro Cova, Patrizio Pagano, Alessandro Notarpietro and Massimiliano Pisani

Number 1156 - December 2017

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Ines Buono, Marco Casiraghi, Valentina Aprigliano, Nicola Branzoli, Francesco Caprioli, Emanuele Ciani, Vincenzo Cuciniello, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valerio Nispi Landi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

SECULAR STAGNATION, R&D, PUBLIC INVESTMENT AND MONETARY POLICY: A GLOBAL-MODEL PERSPECTIVE

by Pietro Cova*, Patrizio Pagano[§], Alessandro Notarpietro* and Massimiliano Pisani*

Abstract

We evaluate the global macroeconomic effects of fiscal and monetary policy measures to counterbalance secular stagnation by simulating a five-region New Keynesian model of the world economy, calibrated to the United States (US), the euro area (EA), Japan (JP), China (CH), and the rest of the world (RW). The model includes investment in research and development (R&D) as a factor that affects global growth. Our main findings are as follows. First, a negative efficiency shock to R&D in the main advanced economies partially replicates the observed slowdown in long-term global growth and the decrease in interest rates. Second, in the medium- and long-term, the increase in US public investment favours global growth; in the short-term, it stimulates US economic activity but reduces foreign activity. Third, in the uS an accommodative monetary stance, which provokes the crowding-in effect, amplifies the short-term macroeconomic effectiveness of public investment, without inducing additional negative spillovers. Fourth, EA, JP, and CH, by simultaneously increasing public investment and adopting an accommodative monetary policy, counterbalance US short-run negative spillovers and further enhance long-term world growth.

JEL Classification: E43; E44; E52; E58

Keywords: DSGE models, secular stagnation, open-economy macroeconomics, public investment, monetary policy.

1. Introduction	5
2. The model	
2.1 Overview	10
2.2 Firms' production function	11
2.3 R&D accumulation, long-run growth and interest rate	14
2.4 Public capital	14
2.5 Monetary authority	
2.6 Equilibrium	
2.7 Calibration	17
3. Simulated scenarios	18
4. Results	19
4.1 Secular stagnation	19
4.2 Increase in US public investment under alternative assumptions	
on the monetary policy stance	
4.3 Simultaneous increase in US, EA, JP, and CH public investment	
4.4 Sensitivity analysis	25
5. Conclusions	
References	27
Tables and Figures	30

Contents

§ World Bank.

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research.

...there is much we can still do to reverse the aggregate productivity slowdown and dispel pessimism about our future. Mario Draghi, President of the ECB.¹

1 Introduction²

The relatively slow recovery of the U.S. (US) economy from the recent financial crisis has resurrected interest in Hansen's secular stagnation hypothesis, i.e, the industrialized economies show an increasing propensity to save and a decreasing propensity to invest, which result in excess savings and a depressed demand, determining low economic growth, inflation and interest rates.³

This view has been brought back to center stage by Summers (2013, 2014, 2015a and 2015b), who has emphasized one version of the hypothesis, based on the ageing of population (which has increased propensity to save) and the decline in the relative price of capital goods induced by technological innovation (which, despite having resulted in a substitution of capital for labour, may have lowered investment expenditure in nominal terms).⁴

A second version of the hypothesis, put forward notably by Gordon (2015), focuses on the supply side and in particular on the rate of growth of productivity, i.e., the economy's potential output for a given amount of available human and material resources employed in the production process. The key argument is that the great inventions that have resulted in massive productivity increases have for the most part already been introduced, so that a return to more moderate growth rates is inevitable.

The secular stagnation phenomenon is not confined to the US economy. Figures

 $^{^{1}}$ Draghi (2017).

²We thank for useful suggestions an anonymous referee, Nicholas Bloom, Ines Buono, and participants at the Banca d'Italia internal workshop on "Secular Stagnation and Financial Cycles" (March 2017), the University of Milan–Polytechnic University of Marche Conference on "Finance and Economic Growth in the Aftermath of the Crisis" (September 2017), the Annual Meeting of the Italian Economic Association (October 2017). The views expressed in this paper are those of the authors alone and should not be attributed to the Bank of Italy, the Eurosystem or the World Bank. Any remaining errors are the sole responsibility of the authors.

³See Hansen (1939).

⁴See also Eichengreen (2015).

1 and 2 report US and euro-area (EA) GDP growth and interest rates over the past four decades, respectively. Both variables clearly display a downward trend. As emphasized by Eggertsson et al. (2016), the average long-term interest rates all over the industrial world are now lower than they were a few years ago, in the immediate aftermath of the crisis, and financial markets suggest that inflation and real interest rates are expected to persistently remain at rather low levels not only in the US but also in Europe and Japan (JP). Therefore, it is relevant to appraise the secular stagnation hypothesis and explore its main policy implications in a global context.

Several policy measures have been suggested to exit from the secular stagnation. Among them is an increase in public infrastructure investment in the US and other main industrialized countries. By stimulating aggregate demand and favoring capital accumulation, possibly in conjunction with an accommodative monetary policy stance, it would help offset the global excess savings with respect to investment and support long-run growth.

We take this "global" approach and evaluate the impact of an increase in public infrastructure investment in the main advanced economies on global growth by simulating a dynamic general equilibrium model. The main contribution of the paper is to fully endogenize long-run growth in a large-scale multi-country New-Keynesian model via R&D accumulation. This allows us to evaluate the effects on both short- and long-run growth of increasing public infrastructure investment. Moreover, we assess the short-run effects of the fiscal expansion under alternative stances of monetary policy.

In the model, the world economy is composed of five blocs, calibrated to the EA, US, China (CH), JP, and the residual "rest of the world" region (RW). Crucially, the model features public infrastructure investment in each region and endogenous accumulation of research and development (R&D) in the tradable sectors of US, EA, JP, and CH (the main global investors in R&D in the data).

Public infrastructure investment is accumulated into public capital, which affects both the demand and the supply sides of the domestic economy. Each region is specialized in the production of final nontradable goods for consumption and investment purposes, and of intermediate tradable and nontradable goods. Both intermediate goods are produced according to a sector-specific Cobb-Douglas technology, that uses private capital, labor (both supplied by domestic households), and public capital. The latter is supplied by the domestic government, financed by raising public debt and lump-sum (non-distortionary) taxes and taken as given by firms when maximizing their profits.

The labor input is affected by a global technology trend, whose growth rate positively depends on the pace of global R&D accumulation. The latter feature is a (rather) novel one for models of this type, and it affects the growth rate of the world economy both in the short and long run (i.e., the world economy follows a longrun balanced growth path, driven by the growth rate of global labor-augmenting technology). R&D is supplied by domestic households, that accumulate it over time and optimally choose the amount of investment. Firms in US, EA, JP, and CH intermediate tradable sectors optimally demand R&D in the domestic perfectly competitive market.

Other features of the model are standard. In each country there are households that maximize utility with respect to consumption and leisure, and firms that maximize profits. Moreover, there are a Taylor-type monetary policy rule, nominal price and wage rigidities, real rigidities (habit in consumption and adjustment costs on investment), and a number of sources of real exchange rate fluctuations, i.e., home bias, local currency pricing, and intermediate nontradable goods.⁵ International financial markets are incomplete, as only a riskless one-period bond, denominated in US dollars, is internationally traded.

The increase in public investment indirectly affects long-run global growth. The higher public capital accumulation makes inputs provided by the private sector, i.e., capital, labor, and R&D, more productive. Thus, firms have an incentive to increase their demand for those inputs, in particular for R&D. The increase in R&D favors the labor-augmenting technological progress and, thus, the long-run (steady-state) growth rate. Moreover, along the long-run balanced growth path of the model there is a single global (natural) interest rate, which holds in all regions and is proportional to the (long-run) growth rate of the economy.

We simulate the following scenarios. Initially, we design a secular stagna-

⁵The monetary policy stance does not affect results in the long run because nominal prices become fully flexible and money neutrality is verified (nominal rigidities hold only in the short run).

tion scenario (first scenario), in which, consistent with a supply-side approach to secular stagnation, the long-run growth rate of the labor-augmenting technology permanently decreases because of a negative shock to the global R&D investment efficiency, i.e., to the capability of converting investment into the (accumulated) stock of R&D. The size of the shock is such that the efficiency of R&D investment is permanently reduced to 90% of its initial level. The shock is in line with estimates provided by Bianchi et al. (2016). The persistent decline in R&D efficiency is documented by Bloom et al. (2017).

Starting from the new (long-run) steady state achieved through the previous simulation ("secular-stagnation steady state") we simulate a permanent increase in US public infrastructure investment by 1% of GDP under two alternative assumptions on the US monetary policy stance: (i) the US monetary authority follows the Taylor rule and, thus, allows for a gradual increase in the policy rate to counteract the expansionary effects of the increase in investment (second scenario); (ii) the authority announces to keep the policy rate constant at the initial steady-state level for two years, thus enacting forward guidance (FG) in the short run (third scenario). Two other scenarios are simulated. In one, public investment is permanently increased in US, EA, JP, and CH by 1% of GDP under a standard monetary policy stance (fourth scenario). In the other, the same increase in public investment occurs under a two-year FG in all the four regions (fifth scenario). The size of the public investment shock is of the same order of magnitude as the reduction in advanced economies' public investment observed in recent decades (see International Monetary Fund 2014).

Finally, we perform a sensitivity analysis. As the effects of the public investment shock are influenced by the relative weight of public capital in private firms' production and by the spillovers of R&D investment across countries, we analyze the sensitivity of the results of the fourth scenario to different calibrations of the corresponding relevant parameters.

All scenarios are simulated under perfect foresight, so that households and firms perfectly anticipate the future path R&D investment efficiency and policy measures.

Our main results are as follows. First, a negative efficiency shock to R&D in the main advanced countries partially replicates the observed slowdown in the long-run

global growth and decrease in interest rates. Second, in the medium and long run the increase in US public investment favors global economic growth; in the short run, it stimulates US economic activity but reduces foreign activity. Third, in the US an accommodative monetary stance, by crowding-in private consumption and investment, amplifies the short-run macroeconomic effectiveness of public investment, without inducing additional negative global spillovers. Fourth, EA, JP, and CH, by simultaneously increasing public investment and adopting an accommodative monetary policy, counterbalance the US short-run negative spillovers and further enhance long-run world economic growth. Finally, results are robust to alternative calibrations of key R&D-related parameters.

Overall, our results clearly speak in favor of coordinating global expansions in public investment.

Some recent contributions have provided a formal treatment of the secular stagnation hypothesis within a dynamic general equilibrium framework. Benigno and Fornaro (2015) build a New Keynesian closed-economy model where pessimistic expectations can permanently reduce productivity growth. Bianchi et al. (2016) build and estimate a dynamic general equilibrium model with endogenous technological progress, where total factor productivity is endogenous and related to R&D and technology adoption rates. Eggertsson et al. (2017) build a closed-economy, overlapping generation model where the zero lower bound is chronically binding. They focus on the role of population growth in reducing equilbrium interest rates. We build on the formalization of R&D as in Bianchi et al. (2016). Different from the above-mentioned contributions, we fully endogenize long-run growth in a large-scale multi-country New-Keynesian model via R&D accumulation.

The paper is organized as follows. Section 2 briefly describes the model, its equilibrium, and the calibration. Section 3 describes the simulated scenarios. Section 4 reports the results. Section 5 concludes.

2 The model

We first provide an overview of the model. Subsequently, we illustrate the crucial features for the simulations. Finally, we report the calibration.

2.1 Overview

We build and simulate a five-region New Keynesian dynamic general equilibrium model of the world economy, calibrated to US, EA, JP, CH, and RW.⁶ In each country households consume, invest in physical capital, R&D (in the case of US, EA, JP, and CH), riskless one-period bonds, and supply labor. One bond is denominated in domestic currency and is traded domestically; a US dollar-denominated bond also exists, that is traded internationally. The domestic-currency bond pays the monetary policy rate set by the domestic central bank. The internationally traded bond pays the US monetary policy rate. The related Euler equations imply that a forward-looking uncovered interest parity condition holds, linking the interest rate differential to the expected depreciation of the currency vis-à-vis the US dollar.⁷

Consumption and investment consist of final nontradable goods, which result from combining constant-elasticity-of-substitution (CES) bundles of intermediate tradable and nontradable goods. The former are domestically produced or imported. Households supply differentiated labor services to domestic firms and act as wage setters in monopolistically competitive labor markets by charging a mark-up over their marginal rate of substitution between consumption and leisure. Finally, households own domestic firms.

On the production side, there are perfectly competitive firms that produce final nontradable goods (consumption and investment goods) and monopolistic firms that produce intermediate goods. The final goods are sold domestically and are produced by combining all available intermediate goods, using a CES production function. The resulting bundles may have different composition.

The model has two rather novel features. First, it allows for public investment in infrastructure in each region. Second, following Bianchi et al. (2016), it allows for endogenous accumulation of R&D. Specifically, both intermediate tradable and nontradable goods are produced according to a sector-specific Cobb-Douglas tech-

 $^{^{6}}$ In each region, size refers to the overall population and to the number of firms operating in each sector. For details on the model equations, see Pesenti (2008) and appendix in Cova et al. (2016).

⁷We make the assumption of a cashless economy, thus we do not consider utility maximization with respect to money demand.

nology, that uses private capital, labor (both supplied by domestic households), and public capital (firms take the public capital stock as exogenously given when maximizing their profits). The labor input is subject to a global technological trend, which positively depends on the accumulated stock of (global) R&D. Firms in the US, EA, JP, and CH intermediate tradable sectors optimally demand R&D in the domestic perfectly competitive market. R&D is supplied by domestic households that accumulate it over time and optimally choose the amount of investment. Crucially, we assume the existence of an "efficiency" shock that affects the accumulation process of R&D.⁸ Moreover, there are R&D spillovers to the other sectors and to other countries.

Finally, in each country there is a Taylor-type monetary policy rule. We also include adjustment costs on real and nominal variables, ensuring that consumption, production, and prices react in a gradual way to a shock. On the real side, habits and quadratic costs delay the adjustment of households consumption and investment (in private capital and in R&D), respectively. On the nominal side, quadratic costs make wages and prices sticky.⁹

In what follows we report the main equations associated with US R&D accumulation, public investment and monetary policy. Similar equations hold in the other regions but RW, which by assumption does not invest in R&D.

2.2 Firms' production function

The production function of the generic firm f^{US} in the US intermediate tradable sector is

$$Y_{T,t}^{US}(f^{US}) = \left(K_{T,t}^{US,P}(f^{US})\right)^{\alpha_{1T}} \left(TREND_{t}^{US}(f) L_{T,t}^{US}(f^{US})\right)^{\alpha_{2T}} \left(K_{t-1}^{US,G}\right)^{1-\alpha_{1T}-\alpha_{2T}}$$
(1)

where $K_{T,t}^{US,P}(f^{US})$ is the demand for private capital, $K_{t-1}^{US,G}$ the public capital, and $L_{T,t}^{US}(f^{US})$ the demand for labor. The parameters $0 < \alpha_{1T}, \alpha_{2T} < 1, \alpha_{1T} + \alpha_{2T} < 1$, are the weights on private capital and labor, respectively. The labor-augmenting technology shock specific to the generic US firm f, $TREND^{US}(f^{US})$, is defined as

 $^{^8 \}mathrm{See}$ Anzoategui et al. (2016).

⁹See Rotemberg (1982).

$$TREND_{t}^{US}(f^{US}) = \left(\left(R\&D_{t}^{US}(f^{US}) \right)^{\eta^{US}} \left(R\&D_{t}^{US} \right)^{1-\eta^{US}} \right)^{\gamma^{US}} \times (R\&D_{t}^{EA})^{\gamma^{EA}} \times \left(R\&D_{t}^{JP} \right)^{\gamma^{JP}} \times \left(R\&D_{t}^{CH} \right)^{1-\gamma^{US}-\gamma^{EA}-\gamma^{JP}}, \forall f^{US},$$

where $R\&D^{US}(f^{US})$ is the US firm f^{US} 's demand for the stock of R&D (accumulated by domestic households), while the aggregate stock of R&D in the US, EA, JP, and CH are respectively

$$R\&D^{US} = \int_0^{n^{US}} R\&D_t^{US}\left(f^{US}\right) df^{US} \tag{3}$$

$$R\&D^{EA} = \int_{n^{US}}^{n^{US}+n^{EA}} R\&D_t^{EA} \left(f^{EA}\right) df^{EA}$$

$$\tag{4}$$

$$R\&D^{JP} = \int_{n^{US}+n^{EA}+n^{JP}}^{n^{US}+n^{EA}+n^{JP}} R\&D_t^{JP}\left(f^{JP}\right) df^{JP}$$
(5)

$$R\&D^{CH} = \int_{n^{US}+n^{EA}+n^{JP}}^{n^{US}+n^{EA}+n^{JP}+n^{CH}} R\&D_t^{CH} \left(f^{CH}\right) df^{CH}$$
(6)

(7)

where $0 < n^{US}, < n^{EA}, < n^{JP}, < n^{CH} < 1$ is the number of firms in the US, EA, JP, and CH tradable sectors, respectively, 1 is the size of the world economy and $n^{US} + n^{EA} + n^{JP} + n^{CH} < 1$.

The $TREND^{US}(f^{US})$ is positively affected by the stock of R&D optimally chosen by the generic firms f in the US, EA, JP, and CH intermediate tradable sectors. When choosing the optimal R&D(f), the generic firm f^{US} takes into account its direct contribution to $TREND^{US}(f^{US})$ (measured by the parameter η , $0 < \eta < 1$). The parameters $\gamma's$ measure the elasticity of $TREND^{US}(f)$ with respect to country-specific R&D ($0 < \gamma^{US}, \gamma^{EA}, \gamma^{JP} < 1, \gamma^{US} + \gamma^{EA} + \gamma^{JP} < 1$).

The generic US firm f optimally demands capital, labor, and R&D (all of them are supplied by domestic households), taking as given prices, the stock of public capital (accumulated by domestic government), the R&D accumulated by other (domestic and foreign) individual firms, and the aggregate R&D in each domestic and foreign sector.¹⁰

A similar trend holds for every firm in EA, JP, and CH, and RW, respectively

$$TREND_{t}^{EA}(f^{EA}) = \left(\left(R\&D_{t}^{EA}(f^{EA}) \right)^{\eta^{EA}} \left(R\&D_{t}^{EA} \right)^{1-\eta^{EA}} \right)^{\gamma^{EA}} \times (8) \\ \left(R\&D_{t}^{US} \right)^{\gamma^{US}} \times \left(R\&D_{t}^{JP} \right)^{\gamma^{JP}} \times \left(R\&D_{t}^{CH} \right)^{1-\gamma^{US}-\gamma^{EA}-\gamma^{JP}}, \\ TREND_{t}^{JP}(f^{JP}) = \left(\left(R\&D_{t}^{JP}(f^{JP}) \right)^{\eta^{JP}} \left(R\&D_{t}^{LP} \right)^{1-\eta^{JP}} \right)^{\gamma^{JP}} \times \\ \left(R\&D_{t}^{US} \right)^{\gamma^{US}} \times \left(R\&D_{t}^{EA} \right)^{\gamma^{EA}} \times \left(R\&D_{t}^{CH} \right)^{1-\gamma^{US}-\gamma^{EA}-\gamma^{JP}}, \\ TREND_{t}^{CH}(f^{CH}) = \left(\left(R\&D_{t}^{CH}(f^{CH}) \right)^{\eta^{CH}} \left(R\&D_{t}^{CH} \right)^{1-\eta^{CH}} \right)^{1-\gamma^{US}-\gamma^{EA}-\gamma^{JP}} \times \\ \left(R\&D_{t}^{US} \right)^{\gamma^{US}} \times \left(R\&D_{t}^{EA} \right)^{\gamma^{EA}} \times \left(R\&D_{t}^{JP} \right)^{\gamma^{JP}}, \\ TREND_{t}^{RW}(f^{RW}) = \left(R\&D_{t}^{CH} \right)^{1-\gamma^{US}-\gamma^{EA}-\gamma^{JP}} \times \\ \times \left(R\&D_{t}^{US} \right)^{\gamma^{US}} \times \left(R\&D_{t}^{EA} \right)^{\gamma^{EA}} \times \left(R\&D_{t}^{JP} \right)^{\gamma^{JP}}. \\ \end{array}$$

As we consider a symmetric equilibrium, in which all firms belonging to the same sector make the same choices, $TREND_t^{US}(f^{US})$ will end up being the same for every US firm. The same is true for every firm in EA, JP, and CH. Thus, $TREND_t^{US}(f^{US})$ will be equal to all trends in other regions. This implies that there is a (common) global trend of labor-augmenting technology shock $TREND_t^{world}$ (i.e., in the symmetric equilibrium the trend is common across all firms producing intermediate tradable and nontradable goods in all regions of the global economy, $TREND_t^{US} = TREND_t^{EA} = TREND_t^{JP} = TREND_t^{CH} = TREND_t^{RW} = TREND_t^{world}$).

Firms in the nontradable sector demand physical capital and labor supplied by domestic households, and take public capital and the (global common) laboraugmenting technology as given (they do not invest in R&D).

¹⁰Firms do not demand public capital and there is no price or tariff paid for its use.

2.3 R&D accumulation, long-run growth and interest rate

Following Bianchi et al. (2016), the US R&D is accumulated by the generic US household i according to

$$R\&D_{t}(i) = (1 - \delta_{R\&D}) R\&D_{t-1}(i) + Z_{I_{R\&D,t}} \left(1 - \frac{\psi_{R\&D}}{2} \left(\frac{I_{R\&D,t}}{I_{R\&D,t-1}} - gr_{t}\right)\right)^{2} I_{R\&D,t}(i)$$
(9)

where we have dropped the "US" superscript for simplicity. The parameter $0 < \delta_{R\&D} < 1$ is the depreciation rate; $\psi_{R\&D} > 0$ is a parameter measuring investment adjustment costs; $I_{R\&D,t}$ is the investment in R&D (whose composition is assumed to be the same as that of private consumption); $Z_{I_{R\&D,t}}$ represents the shock to the marginal efficiency of R&D investment, and gr_t is the gross growth rate of the global labor-augmenting technology trend,

$$gr_t \equiv \frac{TREND_t^{world}}{TREND_{t-1}^{world}}.$$
(10)

Finally, along the long-run balanced growth path the global real (natural) interest rate RR is pinned down by the growth rate gr, endogenously determined by R&D investment, and the households' subjective discount factor $< \beta < 1$,

$$RR = \frac{gr}{\beta}.$$
 (11)

2.4 Public capital

The US fiscal authority exogenously decides the amount of investment in infrastructure and, thus, the accumulation of public capital, $K_{G,t}$, according to

$$K_{G,t} = (1 - \delta_G) K_{G,t-1} + I_{G,t}, \tag{12}$$

where $0 < \delta_G < 1$ is the depreciation rate, and $I_{G,t}$ is public investment.¹¹

¹¹We do not explicitly consider the possibility that public investment takes time to accumulate into physical capital. For the public capital projects with delay between the authorization of a government spending plan and the completion of an investment project, see Kydland and Prescott (1982) and Leeper et al. (2010).

The government budget constraint is

$$B_{G,t} - B_{G,t-1}R_{t-1} \le P_{N,t}C_{G,t} + P_t I_{G,t} - TAX_t,$$
(13)

where $B_{G,t} > 0$ is public debt, which is financed by a one-period nominal bond issued in the domestic bond market, paying the (gross) monetary policy interest rate R_t . The variable $C_{G,t}$ represents government purchases of goods and services, while $TAX_t > 0$ (< 0) are lump-sum taxes (transfers) to households. Consistent with the empirical evidence, $C_{G,t}$ is fully biased towards the nontradable intermediate good. Therefore, it is multiplied by the corresponding price index $P_{N,t}$.¹² Investment in public capital $I_{G,t}$ is assumed to have the same composition as private consumption, in line with the existing literature. Thus, it is pre-multiplied by the consumption price deflator P_t .

The government follows a fiscal rule defined on lump-sum taxes to bring the public debt as a % of domestic GDP, $b_G > 0$, in line with its long-run (steady-state) target \bar{b}_G and to stabilize its rate of change.¹³

The rule is

$$\frac{TAX_t}{TAX_{t-1}} = \left(\frac{b_{G,t}}{\bar{b}_G}\right)^{\phi_1} \left(\frac{b_{G,t}}{b_{G,t-1}}\right)^{\phi_2},\tag{15}$$

where parameters ϕ_1, ϕ_2 are greater than zero, calling for a increase (reduction) in lump-sum taxes whenever the current-period public debt (as a ratio to GDP) is above (below) the target and the previous-period public debt, respectively. We choose lump-sum taxes to stabilize public finance as they are non-distortionary and, thus, allow for a "clean" evaluation of the macroeconomic effects of public investment.

$$GDP_{t} = P_{t}C_{t} + P_{t}^{I}I_{t} + P_{t}I_{R\&D,t} + P_{t}I_{G,t} + P_{N,t}C_{G,t} + P_{t}^{EXP}EXP_{t} - P_{t}^{IMP}IMP_{t},$$
(14)

 $^{^{12}}$ See Corsetti and Mueller (2006).

 $^{^{13}}$ The definition of nominal GDP is

where P_t , is the price of private consumption, public investment, and investment in R&D, given that we assume that public investment and R&D investment bundles have the same composition as private consumption. P_t^I , $P_{N,t}$, P_t^{EXP} , P_t^{IMP} are prices of private investment in physical capital, public consumption, exports, and imports, respectively.

2.5 Monetary authority

In each country the monetary authority sets the policy rate R_t according to a Taylor rule of the form

$$\left(\frac{R_t}{\bar{R}}\right)^4 = \left(\frac{R_{t-1}}{\bar{R}}\right)^{4\rho_R} \left(\frac{\Pi_{t,t-3}}{\bar{\Pi}^4}\right)^{(1-\rho_R)\rho_\pi} \left(\frac{GDP_t}{GDP_{t-1}}\frac{1}{gr_t}\right)^{(1-\rho_R)\rho_{GDP}}.$$
 (16)

The parameter ρ_R (0 < ρ_R < 1) captures the inertia in interest-rate setting, while the term \bar{R} represents the steady-state gross nominal policy rate. The parameters ρ_{π} and ρ_{GDP} are respectively the weights of yearly CPI inflation rate $\Pi_{t,t-3} \equiv P_{C,t}/P_{C,t-4}$ (in deviation from the long-run steady-state target $\bar{\Pi}^4$) and the gross growth rate of the stationary (de-trended) component of GDP, which can be expressed as

$$\frac{GDP_t}{GDP_{t-1}}\frac{1}{gr_t} \equiv \frac{GDP_t/TREND_t^{world}}{GDP_{t-1}/TREND_{t-1}^{world}}.$$
(17)

In some scenarios the central bank is assumed to keep the policy rate constant at its (initial) steady-state level for two years (FG).

2.6 Equilibrium

In each country the initial asset positions, preferences, and budget constraints are the same for all households and for all firms belonging to the same sector. Moreover, profits from ownership of domestic monopolistically competitive firms are equally shared among households. Thus, in each country we have a representative household and a representative firm for each sector (final nontradables, intermediate tradables, and intermediate nontradables). The implied symmetric equilibrium is a sequence of allocations and prices such that, given initial conditions and shocks, households and firms satisfy their corresponding first order conditions, the monetary rules, fiscal rules, and government budget constraints hold, and all markets clear.

2.7 Calibration

Tables 1 to 5 report the (quarterly) calibration of the model parameters, which are set to match the empirical evidence on the pre-secular stagnation period (1970s-80s) and in line with the existing literature.

Table 1 shows the preference and technology parameters. Preferences are the same across households of different regions. The intertemporal elasticity of substitution is set equal to 1.0, the habit parameter to 0.6, and the Frisch elasticity to 0.50. We further assume a depreciation rate of physical and R&D capital to 0.025, consistently with an annual depreciation rate of 10%.

As to final goods, the degree of substitutability between domestic and imported tradables is higher than that between tradables and nontradables (2.5 vs 0.5).

Concerning the parameters related to R&D technology, we set the elasticity of $TREND^{world}$ to regional R&D investment, γ , to 0.30 in EA, 0.40 in US, 0.10 in CH and 0.20 in JP. The contribution of individual firms' R&D investment to $TREND^{world}$ (η) is set to 0.23 in the EA, 0.18 in the US, 0.42 in CH, and 0.35 in JP. These parameters are chosen so that the steady-state ratio of R&D investment to GDP matches aggregate data (see Table 5).

Table 2 reports real and nominal rigidities. For real rigidities, parameters of the adjustment costs on investment changes are set to 4.5 in all countries, both for investment in physical capital and for investment in R&D. For nominal rigidities, we set the adjustment costs for wages to 600; for prices of domestic tradable and nontradable goods, to 600; for prices of imported goods, to 6.00.¹⁴

Table 3 shows price and wage markup values. We identify the intermediate nontradable and tradable sectors in the model with the services and manufacturing sectors in the data, respectively. In each region the markup in the nontradable sector is assumed to be higher than that in the tradable sector and in the labor market, where it is assumed to be equal across sectors.¹⁵

¹⁴The value 600 for quadratic adjustment costs in prices is roughly equivalent to a fourquarter contract length under Calvo-style pricing, as highlighted, among others, by Faruquee et al. (2007). The value 6 for import price adjustment costs is consistent with a relatively quick pass-through of the nominal exchange rate into import prices.

¹⁵Our values are in line with other existing similar studies, such as Bayoumi et al. (2004), Faruqee et al. (2007), Everaert and Schule (2008). Many, if not all, of these studies refer to Jean and Nicoletti (2002) and Oliveira Martins and Scarpetta (1999) for estimates of markups.

Table 4 reports the parameters of the policy rules. For monetary policy rules, the interest rate reacts to the its lagged value (inertial component of the monetary policy), inflation and output growth (see equation 16). For fiscal policy, the parameters governing the speed of adjustment of public debt are assumed equal across countries. Specifically, we set $\phi_1 = 2.01$ and $\phi_2 = 10.01$ (see equation 15).

Table 5 shows the great ratios for the five regions. The weight of domestic tradable goods in the consumption and investment tradable baskets is different across countries, to match multilateral import-to-GDP ratios.

Table 6 reports the trade matrix. We then set the weights of bilateral imports to match it.¹⁶ It is interesting to note that trade with the RW clearly dominates trade patterns for all the other regions.

The implied global growth rate is 2.7%, in line with the evidence reported in Figure 1 and the one provided by the Conference Board (2015) on the growth rate of global TFP.

Finally, we set the discount factor so that, given the steady-state growth rate of the wordwide, the (pre-shock) steady-state annualized real interest rate is 4.8%, in line with the evidence reported in Figure 2.

3 Simulated scenarios

We initially design a secular-stagnation scenario, in which the long-run growth rate of the labor-augmenting technology permanently decreases. Specifically, we simulate a negative shock, $Z_{I_{R\&D}}$, to the global R&D investment efficiency, i.e., to the capability of converting investment into (accumulated) stock of R&D (see equation 9). The size of the shock is such that the efficiency is permanently reduced to 90% of its initial level. The shock is in line with estimates provided by Bianchi et al. (2016). The persistent decline in R&D efficiency is documented, among others, by Bloom et al. (2017).

On top of the secular stagnation, i.e. starting from the new steady state characterized by lower growth and lower interest rates, we simulate a permanent increase

¹⁶We rely on the United Nations' Commodity Trade Statistics (COMTRADE) data on each region's imports of consumer and capital goods, to derive the matrix delineating the pattern and composition of trade for all regions' exports and imports.

in US public infrastructure investment by 1% of GDP under two alternative assumptions on the US monetary policy stance: (i) the US monetary authority follows the Taylor rule (see equation 16) and, thus, allows for a gradual increase in the policy rate to counteract the expansionary effects of the increase in investment (second scenario); (ii) alternatively it announces to keep the policy rate constant at the initial level for two years, thus enacting FG in the short run (third scenario). Two other scenarios are simulated. In one, public investment is permanently increased in US, EA, JP, and CH under a standard monetary policy stance (fourth scenario). In the other, the same increase in public investment occurs under a twoyear FG in all the four regions (fifth scenario). The size of the public investment shock, equal to +1% of GDP, is of the same order of magnitude as the reduction in advanced economies' public investment observed in recent decades.¹⁷ Finally, we perform a sensitivity analysis. As the effects of the public investment shock are influenced by the relative weight of public capital in private firms' production and by the spillovers of R&D investment across countries, we analyze the sensitivity of the results of the fourth scenario to variations in the corresponding relevant parameters.

All scenarios are simulated under perfect foresight, so households and firms perfectly anticipate the future path of R&D investment efficiency and policy measures.

4 Results

4.1 Secular stagnation

We first evaluate the macroeconomic effects of permanently reducing the growth rate of the global labor-augmenting technology shock.

Figures 3 and 4 present the responses of the main US macroeconomic variables to the negative shock to R&D accumulation.¹⁸

As reported in Figure 3, because of the lower R&D investment efficiency, US

¹⁷See International Monetary Fund (2014).

¹⁸In the charts we report the first 80 quarters to show long-run responses. Alternatively, 20 quarters are reported when the emphasis is on the short-run effects, typically when the accommodative monetary policy stance is considered.

firms decrease the growth rate of R&D investment relative to the before-shock long-run growth rate (i.e, the before-shock steady-state balanced growth path). Consistent with that, the global technology growth permanently decreases, implying that the nominal interest rate permanently declines as well. The decline is gradual, consistent with the inertial term in the monetary policy rule. There is an initial increase in the relative price of investment in (private) physical capital, because households substitute investment in physical capital (see below) for investment in less efficient R&D. The initial increase is followed by a permanent, mild decline.

The GDP growth initially undershoots its new lower long-run value, because prices are sticky in the short run and the economy adjusts mainly through changes in the quantities (see Figure 4). Inflation initially decreases and, thereafter, gradually returns to its initial baseline level. Similarly, all of GDP's components undershoot the long-run growth rate. Consumption growth sharply declines on impact, in line with the increase in the ex-ante real interest rate (not reported). The lower consumption growth makes resources available for higher investment in physical capital, whose growth rate initially increases and, after around eight quarters, decreases below the baseline. Exports growth initially increases, favored by the rise in investment in other countries (see below). Imports growth persistently decreases, consistent with the lower growth in the US aggregate demand. Hours worked initially decline, given the initial drop in labor-augmenting technology growth. Thereafter, they increase, in line with the (partial) recovery of the technology trend. The real wage permanently increases. In the short run it barely moves (because of nominal wage and price rigidities). In the long run it stabilizes at a new higher level, because firms augment their demand for labor, to compensate for labor productivity increasing at a slower pace.

Figure 5 depicts the responses of other regions' variables. In every region, the growth rate of GDP falls markedly in the short run. As in the case of the US, that decline is associated with a large decrease in consumption growth, which frees resources for investment growth; the latter increases in the short run to limit, via physical capital accumulation, the decrease in output growth. Hours worked initially decrease and thereafter increase, when the labor productivity is favored by the temporary larger stock of physical capital. Exports growth initially increases in

the EA and JP, while it decreases in CH and RW. To the opposite, import growth increases in CH and RW, and decreases in EA and JP. Consistent with the paths of exports and imports, the EA and JP currencies appreciate in real terms vis-à-vis the US dollar less than the CH and RW currencies do. The mechanism behind the different depreciations reflects differences in R&D. Specifically, the negative R&D shock affects US, EA, JP more than CH, given the chosen calibration (R&D has a lower weight in CH production), while RW does not invest in R&D by assumption. Thus, GDP growth decreases by relatively more in the US, EA, and JP. This favors in the medium run a slightly larger decrease in their policy rates and, thus, the depreciation of their currencies vis-à-vis the CH and RW currencies, that benefits US, EA and JP exports. The reduction in the growth rates induces lower inflation in all regions. Finally, the relative price of investment decreases relatively more in CH and RW because of the appreciation of their exchange rates and the large import content of the investment bundles.

The slower R&D accumulation permanently reduces the growth of technology, which determines the long-run growth rate of the world economy. Thus, the latter converges to a new lower long-run balanced growth path, in which the annualized growth rate is 1.8% (from 2.7%, as reported in Table 7). The global interest rate decreases from 4.8% to 3.8%.

Overall, the negative shock to R&D allows us to replicate the main stylized facts associated with the secular stagnation hypothesis, i.e., the permanent slowdown in global economic activity and the permanent reduction in the global interest rate, in line with the evidence reported in Figures 1 and 2. The associated decline in the trend growth of global labor-augmenting technology matches the estimates by Conference Board (2015).

4.2 Increase in US public investment under alternative assumptions on the monetary policy stance

Figures 6 and 7 present the responses of the main US variables when the US fiscal authority permanently increases public investment by 1% of GDP starting from the secular-stagnation steady state (the new baseline), which the world economy achieved at the end of the previous simulation (i.e., after the negative shock to R&D

accumulation). Public investment in the other regions is instead kept constant at its baseline level. The figures report results obtained when the increase in US public investment is either accompanied by an immediate increase in the US monetary policy rate or, alternatively, by a 2-year FG.

The increase in the US public investment induces firms to increase R&D accumulation (see Figure 6). The effects are more pronounced in the case of 2-year FG, as the larger fall in the short-term real interest rate (not reported) fosters higher investment in R&D on impact. The latter favors a permanent increase in the labor-augmenting technology. In the long run the interest rate increases by approximately 0.2 percentage points.

The US economy benefits from the fiscal and monetary policy measures in the short run (see Figure 7). US GDP growth rate increases. The public deficit-to-GDP ratio initially increases, reflecting the increase in public expenditure. The public debt-to-GDP ratio increases more slowly. Both the deficit and the debt-to-GDP ratio return to their initial levels after about five years, as the fiscal rule (15) is always at work. Hence, there are no long-run variations in the debt-to-GDP ratio, since lump-sum taxes are adjusted to ensure that the target is met. In the case of a 2-year FG the larger expansionary effect on GDP implies an initial small decrease in the debt-to-GDP ratio, which subsequently increases above its initial level in the medium term.

Figures 8-11 report the spillovers on the EA, JP, CH, and RW economies, respectively. Compared to the initial secular stagnation steady state (baseline), the US spillovers are negative in the short run. Focusing, for example, on the EA (Figure 8), GDP growth decreases, because EA households lend to US households in order to finance the additional US growth rate induced by the increase in US public investment. Thus, both EA consumption and investment growth rates initially decrease, so that imports growth initially falls as well. The export growth rate increases, reflecting higher US aggregate demand.

In the case of "accommodative" US monetary policy (i.e. the 2-year FG), the euro appreciates in real terms vis-à-vis the US dollar, implying a slightly negative price-competitiveness effect on the EA tradable goods. However, in the medium run exports growth is (slightly) larger than in the case of standard monetary policy rate, because of the larger US aggregate demand.

Figures 9, 10, 11 show the macroeconomic effects on JP, CH, and RW variables, respectively. Results for JP and CH closely mimic those for the EA. One difference emerges when looking at the RW variables. RW import growth increases in the short run, as its currency appreciates in real terms against the currencies of the other regions. The exchange rate appreciation, by making imports cheaper, induces a positive wealth effect on RW households. Moreover, and different from other regions, the lack of R&D investment induces RW households to substitute consumption for investment growth and hours worked.

Overall, we find that in the short run the mix of permanent increase in US public investment and monetary policy accommodation can partially counterbalance the negative macroeconomic effects of secular stagnation on the world economy.

In the long run the global growth rate registers a larger value than in the (initial) secular-stagnation steady-state. Consistent with that, the global (and common across countries) interest rate increases. As reported in Table 7, the growth rate increases to 2% (from 1.8%), the interest rate to 4.1% (from 3.8%).

4.3 Simultaneous increase in US, EA, JP, and CH public investment

Figures 12-15 report the responses of the US, EA, RW variables when US, EA, JP, and CH simultaneously raise public investment by 1% of GDP.¹⁹ RW public investment is instead constant at its baseline level.

Compared to the case in which the US is the only country to increase public investment (Figure 6) the increase in the global public investment induces firms to increase US R&D at a faster rate in the medium and long run (see Figure 12). Thus, there is a larger permanent increase in the labor-augmenting technology. In the long run the interest rate increases more. Instead, the relative price of private investment increases to a lower extent.

As reported in Figure 13, the US benefits from higher GDP growth in the long run, but it experiences a slower pace of GDP growth in the short run, compared to when it is the sole region to expand public infrastructure spending (compare

¹⁹Results for CH and JP are similar to those of the EA. They are not reported to save on space and are available upon request.

with Figure 7). The result is due to lower global savings available to finance the US expansion. Thus, the US economy relies now more on its own savings, which translates into a stronger fall in US consumption growth in the short run and a correspondingly lower inflation rate. The latter determines a smaller increase of the monetary policy rate in the short run. In the short run the US export growth increases by more than in the case of US-only expansion, favored by the increase in global demand, while US import growth increases by less, because of the lower increase in US aggregate demand. Both the public deficit and the public debt-to-GDP ratio increase to a smaller extent in the medium term, compared to the case of a US-only fiscal stimulus. Both variables are back to their initial levels after five years.

Compared to the case in which public investment increases only in the US, EA GDP growth now rises more in both the short and long run (compare Figure 14 with Figure 8).

In the short run the EA output growth benefits from the increase in global public investment; in the medium and long term, it benefits from the endogenous supply-side effects induced by the additional increase in global R&D, activated by the surge in public infrastructure. Thus, thanks to the fiscal stimulus, R&D accumulation increases not only in the US but also in EA (and in JP and CH, not reported). The EA now exports somewhat less and imports more. The import growth increases because of the increase in public investment. The figure also reports the results for the EA of simultaneously increasing EA, US, JP, and CH public investment under the assumption of 2-year FG. The EA short-run growth rate is magnified by the policy mix. There is less crowding-out of consumption and investment, that favors international trade. The dynamics of the deficit and debt-to-GDP ratios are qualitatively similar to those observed in the US.

In the RW net exports increase to finance the fiscal expansion in the other regions.

Finally, the stronger world aggregate demand also determines on average higher global interest rates both in the short (except for the RW) and in the long run, due to the stronger global growth rate induced by higher R&D accumulation.

As reported in Table 7, in the long run the global growth and interest rate increase to 2.3% and 4.4% from 1.8% and 3.8%, respectively.

Overall, the simultaneous cross-country increase in public investment favors global activity in both short and long run because of the positive effects on R&D. The short-run growth rate of the world economy can be further enhanced by a cross-country accommodative monetary policy stance that accompanies the fiscal stimulus. Our results confirm, as advocated by Summers (2016), that coordination is key for favoring an exit from secular stagnation. If the expansion is solely driven by one country, the US in our case, it will lead in the long run to higher global growth and interest rates, thanks to endogenous growth spillovers, but, in the short run, it will entail negative international spillovers.

4.4 Sensitivity analysis

The results illustrated in the previous section show that a coordinated increase in public investment can effectively increase long-run GDP growth and real interest rates, when advanced economies face a secular stagnation scenario. Those results clearly depend on three crucial parameters: (i) the weight of public capital in firms' production function $(1 - \alpha_{1T} - \alpha_{2T})$, (ii) the contribution of an individual firm to the global trend growth (η) and (iii) the elasticity of the common trend to country-specific R&D. We perform a sensitivity analysis by studying how the long-run effects of the global fiscal stimulus (the one that was shown to be the most effective in stimulating GDP growth) are affected by changes in the three parameters. Since we focus on the long-run effects of the fiscal stimulus, we do not distinguish between the case of Taylor rule and FG and simply report the long-run results.²⁰ Specifically, we first set the relative weight of public capital to 0.2, as opposed to 0.1. We do so by keeping the weight on private capital unchanged, and lowering the weight on labor correspondingly. Second, we increase η by 10% everywhere. Third, we consider an increase in γ^{US} of 0.1pp, from 0.4 to 0.5. As the γ 's must sum up to one, we correspondingly reduce other countries' γ 's by one third each. Table 7 reports the results. A larger weight of public capital in firms' production increases the effectiveness of the fiscal stimulus, both in the case of a US-only increase in public investment and in the case of a coordinated

²⁰For the same reason we do not perform a sensitivity analysis on the R&D investment adjustment cost parameter, $\psi_{R\&D}$, as changes in its calibration only affect the transitional dynamics to the new steady state.

fiscal expansion. Conversely, results are not crucially affected by an increase in the contribution of an individual firm to the global trend, nor by a larger elasticity of the trend to US R&D.

5 Conclusions

This paper has addressed the secular stagnation from a multi-country perspective. We find that unfavorable technology developments can be at the core of the global growth slowdown. We also find that the stagnation can be counterbalanced by appropriate fiscal measures aimed at favoring R&D accumulation. Monetary policy can be a useful complementary lever to favor global growth in the short run, particularly if monetary accommodation is coordinated across countries (regions in our model). Moreover, leaving the burden of enacting an expansionary fiscalmonetary policy mix on one region only results in 'excess savings' in the other regions that hurt their short-term growth prospects and significantly reduce the long-run benefits in terms of higher global growth and interest rates. Expansionary fiscal and monetary policies adopted by all regions can counterbalance the negative short-run spillovers arising from a unilateral fiscal expansion and also enhance long-run world economic growth. Addressing the supply-side headwinds at the core of the secular stagnation with a globally coordinated policy response remains therefore clearly superior in terms of mitigating the output losses.

The paper can be extended along several directions. First, one can allow for the ZLB to constrain monetary policy, thus calling for non-standard measures that directly reduce long-term interest rates. Second, one can consider fiscal measures that would directly affect R&D efficiency (the shock at the basis of secular stagnation in this paper), such as taxes or incentives. We leave these issues for future research.

References

- Anzoategui D., D. Comin, M. Gertler, and J. Martinez (2016). Endogenous Technology Adoption and R&D as Sources of Business Cycle Persistence. NBER Working Papers 22005, National Bureau of Economic Research, Inc.
- [2] Bayoumi, T., D. Laxton and P. Pesenti (2004). Benefits and spillovers of greater competition in Europe: a macroeconomic assessment. Working Paper Series 0341, European Central Bank.
- [3] Benigno, G. and L. Fornaro (2015). Stagnation traps. Working Paper Series 2038, European Central Bank.
- [4] Bianchi, F., H. Kung, and G. Morales (2016). Growth, Slowdowns, and Recoveries, Duke University, mimeo.
- [5] Bloom, N., C. I. Jones, J. Van Reenen, and M. Webb (2017). Are Ideas Getting Harder to Find?, Stanford University, mimeo.
- [6] Conference Board (2015). Productivity brief 2015. Global productivity growth stuck in the slow lane with no signs of recovery in sight.
- [7] Corsetti, G., and G. Mueller (2006). Twin Deficits: Squaring Theory, Evidence and Common Sense. *Economic Policy*. vol. 48, pp. 597–638, October.
- [8] Cova, P., P. Pagano, and M. Pisani (2016). Global macroeconomic effects of exiting from unconventional monetary policy. Temi di discussione (Economic working papers) 1078, Bank of Italy, Economic Research and International Relations Area.
- [9] Draghi, M. (2017). Welcome address by Mario Draghi, President of the ECB, at the joint conference by the ECB and the MIT Lab for Innovation Science and Policy "Fostering Innovation and Entrepreneurship in the Euro area", Frankfurt am Main, 13 March 2017.
- [10] Eichengreen, B. (2015). Secular Stagnation: The Long View. NBER Working Paper No. 20836, January.

- [11] Eggertsson, G., N. R. Mehrotra, and J. Robbins (2017). A Model of Secular Stagnation: Theory and Quantitative Evaluation. NBER Working Paper No. 23093, January.
- [12] Eggertsson, G., N. R. Mehrotra, and L. H. Summers (2016). Secular Stagnation in the Open Economy. American Economic Review: Papers & Proceedings, vol. 106(5), pp. 503–507.
- [13] Everaert, L. and W. Schule (2008). Why It Pays to Synchronize Structural Reforms in the Euro Area Across Markets and Countries. *IMF Staff Papers* vol. 55(2), pp. 356–366.
- [14] Faruquee, H., D. Laxton, and D. Muir (2007). Smooth Landing or Crash? Model-Based Scenarios of Global Current Account Rebalancing. in R. Clarida (ed.) G7 Current Account Imbalances: Sustainability and Adjustment, Chicago, IL: University of Chicago Press.
- [15] Gordon, R. (2015). Secular Stagnation: A Supply-Side View. American Economic Review: Papers & Proceedings vol. 105(5), pp. 54–59.
- [16] Hansen, A. (1939). Economic Progress and Declining Population Growth, American Economic Review, vol. 29, pp. 1–15.
- [17] International Monetary Fund (2014). Is it time for an infrastructure push? The macroeconomic effects of public investment. World Economic Outlook (October), chapter 3.
- [18] Jean, S. and G. Nicoletti (2002). Product Market Regulation and Wage Premia in Europe and North America: An Empirical Investigation. Economics Department Working Paper 419 (Paris: Organisation for Economic Co-operation and Development).
- [19] Kydland, F.E., and E.C. Prescott (1982). Time to build and aggregate fluctuations. *Econometrica*, vol. 50, pp. 1345–1370.
- [20] Leeper, E., T.B. Walker, and S.C.S.Yang (2010). Government Investment and Fiscal Stimulus. *Journal of Monetary Economics*, vol. 57, pp. 1000–1012.

- [21] Oliveira Martins, J. and S. Scarpetta (1999). The Levels And Cyclical Behaviour Of Markups Across Countries And Market Structures. Economics Department Working Paper 213 (Paris: Organisation for Economic Co-operation and Development).
- [22] Pesenti, P. (2008). The Global Economy Model (GEM): Theoretical Framework. *IMF Staff Papers*, vol.55(2), pp.243–284.
- [23] Rotemberg, J. (1982). Monopolistic Price Adjustment and Aggregate Output. *Review of Economic Studies*, Wiley Blackwell, vol. 49(4), pp. 517–31, October.
- [24] Summers, L. H. (2013). "Why Stagnation Might Prove to be the New Normal." The Financial Times.
- [25] Summers, L. H.(2014). "U.S. Economic Prospects: Secular Stagnation, Hysteresis, and the Zero Lower Bound." *Business Economics*, vol. 49(2).
- [26] Summers, L. H. (2015a). "Demand Side Secular Stagnation." American Economics Review, Papers and Proceedings, vol. 105(5), pp. 60–65.
- [27] Summers, L. H. (2015b). "Lower Real Rates, Secular Stagnation and the Future of Stabilization Policy." Central Bank of Chile Annual Conference.
- [28] Summers, L. H. (2016). "The Age of Secular Stagnation." Foreign Affairs, March/April.

	EA	US	CH	JP	RW
Households					
Subjective discount factor	0.995	0.995	0.995	0.995	0.995
Intertemporal elasticity of substitution	1.00	1.00	1.00	1.00	1.00
Habit persistence	0.60	0.60	0.60	0.60	0.60
Inverse of the Frisch elasticity of labor	2.00	2.00	2.00	2.00	2.00
Depreciation rate of capital	0.025	0.025	0.025	0.025	0.025
Depreciation rate of R&D	0.025	0.025	0.025	0.025	0.025
Final consumption goods					
Substitution btw domestic and imp. goods	2.50	2.50	2.50	2.50	2.50
Bias toward domestic goods	0.70	0.70	0.60	0.60	0.90
Substitution btw tradables and nontrad.	0.50	0.50	0.50	0.50	0.50
Bias toward tradable goods	0.50	0.50	0.50	0.50	0.50
Final investment goods					
Substitution btw domestic and imp. goods	2.50	2.50	2.50	2.50	2.50
Bias toward domestic goods	0.60	0.60	0.50	0.70	0.80
Substitution btw tradables and nontrad.	0.50	0.50	0.50	0.50	0.50
Bias toward tradable goods	0.90	0.90	0.90	0.90	0.90
Intermediate tradable goods					
Bias toward private capital	0.30	0.30	0.40	0.30	0.40
Bias toward public capital	0.10	0.10	0.10	0.10	0.10
Intermediate nontradable goods					
Bias toward capital	0.30	0.30	0.30	0.30	0.30
Bias toward public capital	0.10	0.10	0.10	0.10	0.10
R&D technology					
Elasticity of trend to regional R&D	0.30	0.40	0.10	0.20	_
Contribution of firm R&D investment to trend	$0.30 \\ 0.23$	$0.40 \\ 0.18$	$0.10 \\ 0.42$	0.20 0.35	—
Contribution of in in K&D investment to frend	0.25	0.10	0.42	0.55	_

Table 1: Households and firms behavior

Note: EA=euro area; US=United States; CH=China; JP=Japan; RW=Rest of the world.

Real rigidities	
Investment adjustment	4.5
R&D Investment adjustment	4.5
Nominal Rigidities	
Households	
Wage stickiness	600
Manufacturing	
Price stickiness (domestically produced goods)	600
Price stickiness (imported goods)	6
Services	
Price stickiness	600

Note: in each region the corresponding parameter is set equal to the reported value.

Table 3: Gross price and wage markups					
Manufacturing (tradables) price markup	1.20				
Services (non-tradables) price markup	1.30				
Wage markup	1.20				

Note: in each region the corresponding parameter is set equal to the reported value.

Monetary policy rule	
Interest rate inertia	0.87
Interest rate sensitivity to inflation gap	1.70
Interest rate sensitivity to output growth	0.10
Fiscal policy rule	
Lump-sum tax sensitivity to debt gap	2.01
Lump-sum tax sensitivity to debt growth	10.01

Note: in each region the corresponding parameter is set equal to the reported value.

	EA	US	СН	JP	RW
Private consumption	54.0	54.0	51.0	54.0	52.0
Private investment in physical capital	21.0	21.0	25.0	21.0	25.0
Private investment in R&D	1.8	1.8	0.9	1.8	
Public consumption	20.0	20.0	20.0	20.0	20.0
Public investment	3.0	3.0	3.0	3.0	3.0
Imports	15.0	13.0	22.0	18.0	13.0
Consumption goods	9.0	7.0	11.0	13.0	5.0
Investment goods	6.0	6.0	11.0	5.0	8.0
Public debt (% of yearly GDP)	60.0	60.0	60.0	60.0	60.0
Share of world GDP	18.0	20.0	15.0	6.0	41.0

Table 5: Steady state national accounts (% of GDP)

Note: EA=euro area; US=United States; CH=China; JP=Japan; RW=Rest of the world.

	EA	US	CH	JP	RW
Substitution btw consumption imports	2.50	2.50	2.50	2.50	2.50
Imported consumption goods from					
EA		1.3	1.6	1.1	1.7
US	1.9		1.9	1.3	1.3
СН	1.5	1.3		1.9	2.0
JP	0.5	0.4	1.3		0.4
RW	5.0	4.2	6.2	9.0	
Substitution btw investment imports	2.50	2.50	2.50	2.50	2.50
Imported investment goods from					
EA		1.0	1.5	0.4	2.3
US	0.1		1.8	0.5	1.8
СН	1.0	1.0		0.7	2.8
JP	0.3	0.3	1.2		0.6
RW	3.6	3.2	5.7	3.6	
Net foreign assets	0	0	0	0	0

Table 6: International linkages (% of GDP)

-

=

Note: EA=euro area; US=United States; CH=China; JP=Japan; RW=Rest of the world.

Table 7: Global fiscal stimulus

	Bench	mark	Sensitivity					
			High we	eight of K^G	Hig	n η	High	γ^{US}
	GDP	R	GDP	R	GDP	R	GDP	R
(1): Initial steady state	2.7	4.8	2.4	4.4	3.6	5.6	2.8	4.9
(2): Secular stagnation	1.8	3.8	1.6	3.7	2.6	4.7	1.9	4.0
difference (2) - (1)	-0.9	-1.0	-0.8	-0.8	-1.0	-1.0	-0.9	-0.9
(3): $(2) + US$ pub. inv.	2.0	4.1	1.9	3.9	2.8	4.9	2.2	4.2
difference (3) - (2)	0.2	0.2	0.3	0.3	0.2	0.2	0.3	0.2
(4): $(2) + \text{global pub. inv.}$	2.3	4.4	2.3	4.4	3.1	5.2	2.4	4.5
difference (4) - (2)	0.5	0.5	0.7	0.7	0.5	0.5	0.5	0.5

Note: GDP is growth rate of real GDP, in % points. Real interest rate (R) is in level (% points). "High weight of K^{G} ": the weight of public capital is 0.2 (from 0.1). "High η ": the elasticity of technology trend with respect to each individual firm' R&D is multiplied by 1.1. "High γ^{US} ": the elasticity of technology trend with respect to US R&D is increased by 0.1pp and other "regional" elasticities are correspondingly decreased by 1/3 of a pp.


Figure 1: Secular stagnation. EA and US GDP

Note. GDP in real terms. Ten-year centered moving averages of annual growth rates, % points. Sources: St. Louis Fed FRED and Area Wide Model Dataset. EA= euro area; US= United States.



Figure 2: Secular stagnation. EA and US real interest rates

Note. Interest rates on ten-year government bonds, % points. Sources: St. Louis Fed FRED and Area Wide Model Dataset. EA= euro area; US= United States.





Note. Horizontal axis: quarters; vertical axis: R&D investment and labor-augmenting technology growth rates reported as annualized p.p. deviations from the pre-secular-stagnation steady state; nominal interest rate as annualized p.p.; investment relative price as % deviations.



Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the pre-secular-stagnation steady state; CPI inflation as annualized p.p. deviations; hours worked and real wage as % deviations.



Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the pre-secular-stagnation steady state; interest rate as annualized p.p.; CPI inflation as annualized p.p. deviations; investment relative price, real exchange rate, hours worked and real wage as % deviations.





Note. Horizontal axis: quarters; vertical axis: R&D investment and labor-augmenting technology growth rates reported as annualized p.p. deviations from the secular-stagnation steady state; nominal interest rate as annualized p.p.; investment relative price as % deviations.



Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; CPI inflation as annualized p.p. deviations; hours worked and real wage as % deviations; public deficit (ratio to GDP) and debt (ratio to annualized GDP) as p.p. deviations.



Figure 8: Secular stagnation, US public inv. and monetary stance. EA variables

Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; interest rate as annualized p.p. (level); CPI inflation as annualized p.p. deviations; investment relative price, real exchange rate, hours worked and real wage as % deviations.



Figure 9: Secular stagnation, US public inv. and monetary stance. JP variables

Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; interest rate as annualized p.p. (level); CPI inflation as annualized p.p. deviations; investment relative price, real exchange rate, hours worked and real wage as % deviations.



Figure 10: Secular stagnation, US public inv. and monetary stance. CH variables

Note. Horizontal axis: quarters; vertical axis: nominal interest rate as annualized p.p.; real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; interest rate as annualized p.p. (level); CPI inflation as annualized p.p. deviations; investment relative price, real exchange rate, hours worked and real wage as % deviations.



Figure 11: Secular stagnation, US public inv. and monetary stance. RW variables

Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; interest rate as annualized p.p. (level); CPI inflation as annualized p.p. deviations; investment relative price, real exchange rate, hours worked as % deviations.

Figure 12: Secular stagnation, global public inv. and monetary stance. US variables and global technology



Note. Horizontal axis: quarters; vertical axis: R&D investment and labor-augmenting technology growth rates reported as annualized p.p. deviations from the secular-stagnation steady state; interest rate as annualized p.p. (level); investment relative price as % deviations.



Figure 13: Secular stagnation, global public inv. and monetary stance. US variables

Note. Horizontal axis: quarters; vertical axis: real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; CPI inflation as annualized p.p. deviations; hours worked and real wage as % deviations; public deficit (ratio to GDP) and debt (ratio to annualized GDP) as p.p. deviations.



Figure 14: Secular stagnation, global public inv. and monetary stance. EA variables

Note. Horizontal axis: quarters; vertical axis: nominal interest rate as annualized p.p. (level); real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; CPI inflation as annualized p.p. deviations; hours worked and real wage as % deviations; public deficit (ratio to GDP) and debt (ratio to annualized GDP) as p.p. deviations.



Figure 15: Secular stagnation, global public inv. and monetary stance. RW variables

Note. Horizontal axis: quarters; vertical axis: nominal interest rate as annualized p.p. (level); real GDP and its components' growth rates as annualized p.p. deviations from the secular-stagnation steady state; CPI inflation as annualized p.p. deviations; hours worked and investment relative price as % deviations.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1129 *The effects of central bank's verbal guidance: evidence from the ECB*, by Maddalena Galardo and Cinzia Guerrieri (July 2017).
- N. 1130 The Bank of Italy econometric model: an update of the main equations and model elasticities, by Guido Bulligan, Fabio Busetti, Michele Caivano, Pietro Cova, Davide Fantino, Alberto Locarno, Lisa Rodano (July 2017).
- N.1131 Venture capitalists at work: what are the effects on the firms they finance?, by Raffaello Bronzini, Giampaolo Caramellino and Silvia Magri (September 2017).
- N.1132 Low frequency drivers of the real interest rate: a band spectrum regression approach, by Fabio Busetti and Michele Caivano (September 2017).
- N. 1133 *The real effects of relationship lending*, by Ryan Banerjee, Leonardo Gambacorta and Enrico Sette (September 2017).
- N.1134 *Credit demand and supply: a two-way feedback relation*, by Ugo Albertazzi and Lucia Esposito (September 2017).
- N. 1135 Legislators' behaviour and electoral rules: evidence from an Italian reform, by Giuseppe Albanese, Marika Cioffi and Pietro Tommasino (September 2017).
- N.1136 Macroeconomic effects of non-standard monetary policy measures in the euro area: the role of corporate bond purchases, by Anna Bartocci, Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2017).
- N. 1137 On secular stagnation and low interest rates: demography matters, by Giuseppe Ferrero, Marco Gross and Stefano Neri (September 2017).
- N. 1138 Tony Atkinson and his legacy, by Rolf Aaberge, François Bourguignon, Andrea Brandolini, Francisco H. G. Ferreira, Janet C. Gornick, John Hills, Markus Jäntti, Stephen P. Jenkins, Eric Marlier, John Micklewright, Brian Nolan, Thomas Piketty, Walter J. Radermacher, Timothy M. Smeeding, Nicholas H. Stern, Joseph Stiglitz and Holly Sutherland (September 2017).
- N. 1139 *Credit misallocation during the European financial crisis*, by Fabiano Schivardi, Enrico Sette and Guido Tabellini (September 2017).
- N. 1140 Natural rates across the Atlantic, by Andrea Gerali and Stefano Neri (September 2017).
- N.1141 A quantitative analysis of risk premia in the corporate bond market, by Sara Cecchetti (October 2017).
- N. 1142 *Monetary policy in times of debt*, by Mario Pietrunti and Federico M. Signoretti (October 2017).
- N.1143 Capital misallocation and financial development: a sector-level analysis, by Daniela Marconi and Christian Upper (October 2017).
- N. 1144 Leaving your mamma: why so late in Italy?, by Enrica Di Stefano (October 2017).
- N. 1145 A Financial Conditions Index for the CEE economies, by Simone Auer (October 2017).
- N. 1146 Internal capital markets in times of crisis: the benefit of group affiliation in Italy, by Raffaele Santioni, Fabio Schiantarelli and Philip E. Strahan (October 2017).
- N. 1147 *Consistent inference in fixed-effects stochastic frontier models*, by Federico Belotti and Giuseppe Ilardi (October 2017).
- N. 1148 Investment decisions by European firms and financing constraints, by Andrea Mercatanti, Taneli Mäkinen and Andrea Silvestrini (October 2017).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBERTAZZI U., G. ERAMO, L. GAMBACORTA and C. SALLEO, Asymmetric information in securitization: an empirical assessment, Journal of Monetary Economics, v. 71, pp. 33-49, TD No. 796 (February 2011).
- ALESSANDRI P. and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- ANTONIETTI R., R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- BARONE G. and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- BRONZINI R., The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- BUGAMELLI M., S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- BULLIGAN G., M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- BUSETTI F., On detecting end-of-sample instabilities, in S.J. Koopman, N. Shepard (eds.), Unobserved Components and Time Series Econometrics, Oxford, Oxford University Press, **TD No. 881 (September 2012).**
- CESARONI T., *Procyclicality of credit rating systems: how to manage it,* Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- CIARLONE A., *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- CUCINIELLO V. and F. M. SIGNORETTI, *Large bank, loan rate markup and monetary policy,* International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- DE BLASIO G., D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds,* Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, **TD No. 792 (February 2011).**
- DEPALO D., R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- DI CESARE A., A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- FANTINO D., A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- FRATZSCHER M., D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- NOTARPIETRO A. and S. SIVIERO, *Optimal monetary policy rules and house prices: the role of financial frictions*, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, **TD No. 993 (November 2014).**
- RIGGI M. and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**
- TANELI M. and B. OHL, *Information acquisition and learning from prices over the business cycle,* Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **TD No. 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- ESPOSITO L., A. NOBILI and T. ROPELE, *The management of interest rate risk during the crisis: evidence from Italian banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, TD No. 968 (July 2014).

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010)**.
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

FORTHCOMING

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038 (November 2015).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis,* Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, **TD No. 1077 (July 2016).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).